Author’s response to reviews

Title: Analysis of Proliferative Activity in Oral Gingival Epithelium in Immunosuppressive Medication Induced Gingival Overgrowth

Authors:

Sule Bulut (suleb@baskent.edu.tr)
Hilal Uslu (usluhilaladana@yahoo.com)
Handan B Ozdemir (handano@baskent-ankara.edu.tr)
Engin O Bulut (omerenginbulut@gmail.com)

Version: 3 Date: 8 May 2006

Author’s response to reviews:

Dear Editor,

The latest revision was done and according to the comments of second Reviewer the sentence of "Four sites were measured within each sample and then mean epithelial thickness was calculated." was placed in methods section. I wish to receiving your acception message for the publication of our manuscript in your journal. Thank you for your interest once again

Sincerely yours,